Author's response to reviews

Title: Augmentation Of Tibial Plateau Fractures With An Injectable Bone Substitute: CERAMENTTM. Three year follow-up from a prospective study.

Authors:

Riccardo Iundusi (riccardo.iundusi@uniroma2.it)
Elena Gasbarra (elenagasbarra@tiscali.it)
Michele D'Arienzo (mdarienzo@unipa.it)
Andrea Piccioli (piccioli.and@gmail.com)
Umberto Tarantino (umberto.tarantino@uniroma2.it)

Version: 2
Date: 24 December 2014

Author's response to reviews:

Point-by-point description of the requested changes made:

1. in the title page we added complete address of different affiliations, email addresses of all authors and the full postal address of the corresponding author.

2. the text now has double line spacing with line and page numbers